Author's response to reviews

Title: Spontaneous preterm labor is associated with an increase in the proinflammatory signal transducer TLR4 receptor on maternal blood monocytes.

Authors:

Edyta Pawelczyk (pawelczyke@cc.nih.gov)
Bogdan J Nowicki (bnowicki@mmc.edu)
Michael G Izban (mizban@mmc.edu)
Pratap Siddharth (spratap@mmc.edu)
Sashti A. Nupur (Nupur.A.Sashti@tn.gov)
Sanderson Maureen (ssanderson@mmc.edu)
Stella Nowicki (snowicki@mmc.edu)

Version: 3 Date: 14 October 2010

Author's response to reviews: see over
Dear Editor,

Enclosed please find a resubmission of our manuscript entitled “Spontaneous Preterm Labor is Associated with an Increase in the Proinflammatory Signal Transducer TLR4 Receptor on Maternal Blood Monocytes. “

Thank you so much for the acceptance of our manuscript. We addressed the editorial comments and modified our manuscript as suggested.

Major revisions

1. Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name, Initial, Last name' (e.g. Joe F Bloggs)

The list of authors was corrected.

2. Structure: this manuscript does not have the correct structure. Please reformat it to use these headings: Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Authors' information (optional); Acknowledgements (optional); References

The manuscript was reformatted according to correct structure

3. Authors' contributions: Please use the following format as an example, using initials to refer to each author's contribution: AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination.

Authors' contributions were reformatted.

4. Please state that all authors read and approved the final manuscript.

This statement is now included
Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

5. Title: please use sentence case in the title, both in the manuscript file and the submission system. Capitalise only the first word and proper nouns, and do not include full stops/periods. For example, `The Title of My Prestigious Publication: A Scientific Study.' should be changed to `The title of my prestigious publication: a scientific study'

Title was corrected

6. Affiliations: Please remember to include the country in the affiliation details

The country was included

7. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We checked manuscript for typographical errors.

Thanks

BJN